Author’s response to reviews

Title: Correlation between compliance and brace treatment in juvenile and adolescent idiopathic scoliosis. SOSORT 2014 award winner.

Authors:

Angelo G Aulisa (angelogabriele.aulisa@fastwebnet.it)
Marco Giordano (marco.giordano@opbg.net)
Francesco Falciglia (francesco.falciglia@opbg.net)
Emanuele Marzetti (emarzetti@live.com)
Andrea Poscia (andreaposcia83@gmail.com)
Vincenzo Guzzanti (vguzzanti@yahoo.it)

Version: 4 Date: 6 June 2014

Author’s response to reviews: see over
Reviewer #Toru Maruyama

Discretionary Revisions

- I suggest that the definition of incomplete A and B should be written as brace removed for 1(2) month(s) of a year, or 1(2) month(s) per year for clarification.

We have changed in the text, when possible, the definition Incomplete A and B with: “brace removed for 1 month of a year (incomplete A), brace removed for 2 months of a year (incomplete B)”. We have made the same for Incomplete C and D: “brace removed during school hours (incomplete C), and brace worn overnight only (incomplete D)”.